Author's response to reviews

Title: Hip fracture in hospitalized medical patients

Authors:

Antonio Zapatero (azapatero.hflr@salud.madrid.org)
Raquel Barba (raquel.barba@hospitalreyjuancarlos.es)
Jesús Canora (jcanora.hflr@salud.madrid.org)
Juan Emilio Losa (JELosa@fhalcorcon.es)
Susana Plaza (splaza.hsvo@salud.madrid.org)
Jesús San Román (jesus.sanroman@urjc.es)
Javier Marco (javier.marco@elmundo.es)

Version: 9 Date: 18 December 2012

Author's response to reviews: see over
Author’s response to reviews

Title: Hip fracture in hospitalized medical patients

Authors

Antonio Zapatero, (azapatero.hflr@salud.madrid.org)
Raquel Barba, (raquel.barba@hospitalreyjuancarlos.es)
Jesús Canora, (jcanora.hflr@salud.madrid.org)
Juan E Losa, (JELosa@fhalconcon.es)
Susana Plaza, (splaza.hsvo@salud.madrid.org)
Jesús San Roman, (jesus.sanroman@urjc.es)
Javier Marco (javier.marco@elmundo.es)

Version: 8 Date 18 December 2012

Author’s response to reviews: see over
Reviewer's report
Title: Hip fracture in hospitalized medical patients
Version: 8 Date: 4 December 2012
Reviewer: Masayuki Iki

Reviewer's report:
The authors have revised the manuscript according to the comments and suggestions from the reviewer. There are some small errors as follows:

Discretionary Revisions:
1. L149 The authors do not have to add “unadjusted” and “the regression coefficients” should be “the logistic regression coefficients”.

Added “unadjusted” and “the logistic regression coefficients”.

2. Table 3: Too many decimals for Cases/patient•days. The three or four decimal places may be enough. “Patient/days” should be “patient•days”.

Corrected.